Title: Perceived quality of life in obsessive-compulsive disorder: Related factors

Authors:

Beatriz Rodriguez-Salgado (beatriz.rodriguez@hrc.es)
Helen Dolengevich-Segal (edolen@ya.com)
Manuel Arrojo-Romero (manuel.arrojo@hrc.es)
Paola Castelli-Candia (paola.castelli@ucm.es)
Mercedes Navio-Acosta (mercedes.navio@hrc.es)
Maria M Perez-Rodriguez (merperez@yahoo.com)
Jeronimo Saiz-Ruiz (jeronimo.saiz@hrc.es)
Enrique Baca-Garcia (ebacgar2@yahoo.es)

Version: 4 Date: 24 March 2006

Author's response to reviews: see over
Dear editor,

We are submitting the revision of the paper "Perceived quality of life in obsessive-compulsive disorder: Related factors".

Query: “Ethical approval - Please include the name of the body that supplied Ethical approval for your study, with a reference number where appropriate.”

Answer: The name of the body that supplied Ethical approval for our study is the Comité de Ética del Hospital Ramón y Cajal (Ethical Committee of Ramon y Cajal Hospital). We have included a statement in the Methods section (in italics, page 5, paragraph 4):

“The research carried out is in compliance with the Helsinki Declaration. The name of the body which gave approval is the Ethical Committee for Clinical Research (Comité Ético de Investigación Clínica) of Ramon y Cajal Hospital (record of meeting number 118, Research Project 042/03).”

If you need any further information please do not hesitate to contact us.

Yours sincerely,

Beatriz Rodriguez Salgado,
Helen Dolengevich Segal,
Manuel Arrojo Romero,
Paola Castelli Candia,
Mercedes Navio Acosta,
M. Mercedes Perez-Rodriguez,
Jeronimo Saiz Ruiz,
Baca-Garcia, Enrique